Author’s response to reviews

Title: Insulin analogues dosing and costs - comparing real-life daily doses of insulin Detemir and insulin Glargine in type 2 diabetes patients

Authors:

Marie Jakobsen (majk@cowi.dk)
Mette Dalsgaard (medal@mst.dk)
Morten Hørmann (mho@cowi.dk)
Daniel V Møller (dvm@novonordisk.com)

Version: 2 Date: 6 July 2012

Author’s response to reviews:

In the revised version of the manuscript, we have incorporated the following text into the methods section concerning ethics approval:

"As no personal data defined as any information relating to an identified or identifiable patient was obtained in the questionnaires to GPs and specialists (e.g. CPR number (civil identification number) or other personal data which could refer to a single patient), no informed consent was necessary by Danish law. A waiver for ethics approval has been obtained."